CS 1321

Homework #0: Background Questionnaire

Assigned: January 12, 2000 Due: January 12, 2000 Total points: 5

• Your name:

• Your major and year (e.g., “CS, junior”):

• When did you take CS 1320, and with what instructor?

• Are you also taking CS 1323 this term? or have you taken it previously?

• Do you have any suggestions for how to make this class more interactive?

• To give me an idea of how much you remember from CS 1320 — and to start the C++ gears in your brain working again — on the back of this page please define a recursive C++ function that, given parameters double b and non-negative int e, returns b to the e power. Comment your function appropriately. (If you don’t remember enough from CS 1320 to answer this question, just say so. There are no wrong answers to any questions on this page.)